Institut for Matematik og Datalogi Syddansk Universitet

April 26, 2012 JFB DM830 Network Security – F12 – Lecture 5

Lecture, April 23

We finished chapter 3 and covered up through page 132 of chapter 4.

Lecture, April 27

We will finish chapter 4, cover Appendix D in the textbook (see the textbook’s Web page) and begin on SSH in chapter 5.

Lecture, May 2

We will finish chapter 5 and begin on chapter 7.

Discussion section: May 7

There will be presentations on worms from chapter 10 in the textbook and on time-memory trade-offs in cryptanalysis.

1. Problem 4.18.

2. Problem 5.5.